Author's response to reviews

Title: Eosinophilic pneumonia associated with daptomycin: a case report and a review of the literature

Authors:

Andreas S Kalogeropoulos (andkalog@gmail.com)
Sotirios Tsiodras (tsiodras@med.uoa.gr)
Dionysios Loverdos (epathologiki@evaggelismos-hosp.gr)
Panagiotis Fanourgiakis (agoritsa@otenet.gr)
Athanasios Skoutelis (skout@med.upatras.gr)

Version: 3 Date: 21 September 2010

Author's response to reviews: see over
Michael Kidd, MD

Editor-in-Chief, Journal of Medical Case Reports

E-mail address: jmcredtorial@biomedcentral.com

Manuscript title: Eosinophilic Pneumonia Associated with Daptomycin: A Rare Case and Literature Review.

Authorship: Andreas S. Kalogeropoulos, MD, Sotirios Tsiodras¹, MD, MSc, PhD, Dionysios Loverdos, MD, Panagiotis Fanourgiakis, MD, Athanasios Skoutelis, MD

Affiliations: 5th Department of Internal Medicine and Infectious Diseases, “EVANGELISMOS” General Hospital, Athens, Greece

1 4th Academic Department of Internal Medicine and Infectious Diseases, University of Athens Medical School, Attikon University Hospital, Athens, Greece

Dear Dr. Kidd,

We have carefully examined all the suggestions that were raised by the editorial team regarding the manuscript format and we proceeded to all the necessary changes.

In particular the title has been changed to "Eosinophilic pneumonia associated with daptomycin: a case report and a review of the literature" instead of "Eosinophilic Pneumonia Associated With Daptomycin: a Rare Case and Literature Review". Furthermore, in the abstract segment we changed the "Background" to "Introduction" and the "Case summary" to "Case presentation". Moreover, we changed the "Background" to
"Introduction" in the introduction segment, we replaced the authors' full names with initials in the e-mail list and we removed all highlighting from the manuscript. In addition, we have made some other changes, listed below, including:

- In the introduction segment, in the second paragraph, 7th line, we replaced the phrase "pointed at" with the phrase "pointed out".

- In the case presentation segment of the manuscript, in the first paragraph, 2nd line, we replaced the word "presented" with the phrase "was referred".

- In the case presentation segment of the manuscript, in the 4th paragraph, 9th line, we changed the phrase "which was accompanied" to "accompanied".

- In the discussion segment of the manuscript, in the 1st paragraph, 9th line, we changed the word "Furthermore" to "In addition". At the same paragraph, 10th line, we changed the phrase "that in turn" to "which in turn" and in the 11th line we added the word "further".

- In the discussion segment of the manuscript, in the 2nd paragraph, 22nd line, we changed the word "analyses" to "analysis".

I would be more than willing to respond, should you have any questions regarding the aforementioned changes.

Sincerely,

Andreas S. Kalogeropoulos, MD

5th Department of Internal Medicine and Infectious Diseases
“EVANGELISMOS” General Hospital

45-47 Ipsilantou Street

106 76 Kolonaki

Athens, Greece

Fax: 0030-210-7216408

Tel: 0030-210-7201830, 0030-210-7205177

E-mail: andkalog@gmail.com, ankalog@med.uoa.gr